Author's response to reviews

Title: An Unruptured, Thrombosed 10 cm Right Coronary Artery Aneurysm mimicking a Pericardial Cyst

Authors:

Aakash Chauhan (achauhan1000@gmail.com)
Harsha Musunuru (hmusunuru@gmail.com)
Richard L Hallett (xraydoc97@yahoo.com)
Mark M Walsh (markwalshmd@gmail.com)
Szabolcs Szabo (sszabolcs@pol.net)
Walter H Halloran (walterhal@aol.com)

Version: 2 Date: 14 June 2012

Author's response to reviews: see over
Dear Dr. Zamvar,

We thank you for your interest in our paper and welcome the recommendations by the referee. Our adjustments to the manuscript are as follows:

Reviewer's Report:

Discretionary Revisions: This is a very well-written case report and has wonderful illustrations. My only suggestion is to narrow down the illustrations to the most relevant ones. To this, I would consider cutting Figure 3 and Figure 6.

We would like to thank Dr. Miller for his comments. As recommended, we have removed Figures 3 and 6 from the submission. We have changed the labeling of the figures in the text to reflect this change.

We hope that the resultant changes meet the satisfaction of the reviewers. Again, thank you for your interest.

Aakash Chauhan MD,MBA